



© Open Access Limnology and Oceanography Letters 9, 2024, 112–118 © 2023 The Authors. Limnology and Oceanography Letters published by Wiley Periodicals LLC on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/lol2.10369

LETTER

Ice-melt period dominates annual carbon dioxide evasion from clear-water Arctic lakes

J. Karlsson ^(D), ¹* H. A. Verheijen ^(D), ¹ D. A. Seekell ^(D), ¹ D. Vachon ^(D), ¹ M. Klaus^{1,2}

¹Climate Impacts Research Centre, Department of Ecology and Environmental Science, Umeå University, Umeå, Sweden; ²Department of Forest Ecology and Management, Swedish University of Agricultural Sciences, Umeå, Sweden

Scientific Significance Statement

Arctic lakes are thought to release a significant amount of carbon dioxide (CO_2) into the atmosphere. However, current estimates are uncertain because most studies occur during the summer months, and the CO_2 that accumulates in winter and largely evades to the atmosphere within a few days when ice melts in spring is rarely quantified. In a study of 14 Arctic lakes, we show that the ice-melt period dominates (mean 80%) annual CO_2 evasion. Comparisons with previous studies reveal that ice-melt CO_2 evasion is particularly important in clear-water organic carbon-poor lakes, which are abundant in the Arctic, and decreases toward more colored organic carbon-rich lakes. The results stress that overlooking the high share of CO_2 evasion at ice-melt likely underestimates CO_2 release to the atmosphere from Arctic lakes.

Abstract

Current estimates of carbon dioxide (CO₂) evasion from Arctic lakes are highly uncertain because few studies integrate seasonal variability, specifically evasion during spring ice-melt. We quantified annual CO₂ evasion for 14 clear-water Arctic lakes in Northern Sweden through mass balance (ice-melt period) and high-frequency loggers (open-water period). On average, 80% (SD: \pm 18) of annual CO₂ evasion occurred within 10 d following ice-melt. The contribution of the ice-melt period to annual CO₂ evasion was high compared to earlier studies of Arctic lakes (47% \pm 32%). Across all lakes, the proportion of ice-melt : annual CO₂ evasion was negatively related to the dissolved organic carbon concentration and positively related to the mean depth of the lakes. The results emphasize the need for measurements of CO₂ exchange at ice-melt to accurately quantify CO₂ evasion from Arctic lakes.

Lakes emit significant amounts of greenhouse gases into the atmosphere (Raymond et al. 2013). The gas evasion is dominated by carbon dioxide (CO_2) and driven by the input of inorganic and organic carbon (C) from land (Tranvik et al. 2009). High-latitude lakes are abundant (Verpoorter et al. 2014), generally small (Muster et al. 2019), and with

Associate editor: Emily H. Stanley

Author Contribution Statement: JK designed the study with contributions from all authors. HAV and DV conducted the fieldwork. HAV and MK analyzed the data and conducted statistical analysis. JK and HAV wrote the first draft of the paper. All authors provided comments on the paper before submission.

Data Availability Statement: Data and metadata are available at https://zenodo.org/records/10080521.

Additional Supporting Information may be found in the online version of this article.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

^{*}Correspondence: jan.p.karlsson@umu.se

close hydrological connectivity to terrestrial ecosystems, facilitating the input and evasion of land-derived C forms in these lakes (Vachon et al. 2017). The pronounced warming at high latitudes is expected to affect terrestrial C cycling and lateral C export, but also the mineralization and evasion of this C in lakes (Williamson et al. 2008). It is, therefore, important to quantify and understand lake C evasion to accurately assess the contemporary and future Arctic C cycle.

Many lakes show strong temporal variability in atmospheric CO₂ exchange. For northern lakes, especially with long winter and ice-covered periods, CO2 accumulates under ice and is largely released during a short period following icemelt in spring (Striegl and Michmerhuizen 1998; Karlsson et al. 2013). The CO_2 accumulation during winter and subsequent evasion at ice-melt generally increases with lake mean depth (Karlsson et al. 2013; Ducharme-Riel et al. 2015). A synthesis of data reported that ice-melt CO₂ evasion constituted, on average, 17% of annual evasion in northern lakes, with especially high values in the Arctic (34%) compared to boreal (16%) and temperate (22%) lakes (Denfeld et al. 2018). However, this synthesis only included 18 Arctic lakes, and recent studies in clear-water Arctic lakes with low content of dissolved organic C (DOC) reported higher values (58% \pm 32% in 14 Swedish lakes, Verheijen et al. 2022; 51% and 71% in 2 Canadian lakes, Preskienis et al. 2021). The importance of ice-melt evasion could thus be undervalued given the abundance of low DOC lakes (median DOC < 5 mg L^{-1}) in the Arctic compared to other biomes (Sobek et al. 2007), calling for specific investigation of the importance of ice-melt evasion in these types of systems. However, most assessments of CO₂ evasion from lakes, including in the Arctic, neglect seasonal variability in CO₂ fluxes, focusing instead on a limited number of sampling occasions during the summer (Klaus et al. 2019). This implies that current assessments of lake CO_2 evasion are likely underestimates, especially in clear-water Arctic lakes.

The aim of this study was to investigate the ice-melt contribution to annual CO_2 evasion from Arctic clear-water, low DOC lakes. We quantified the annual CO_2 exchange for 14 clear-water lakes in northern Sweden based on dissolved inorganic C (DIC) mass balance during the ice-melt period and high-frequency logging during the open-water period and compared the results to published data to explore general patterns in ice-melt CO_2 evasion across Arctic lakes.

Methods

Study lakes

The 14 lakes are about 250 km north of the Arctic Circle in northern Sweden and range from 1.6 to 11.4 ha in surface area, 2.3 to 20.5 m in maximum depth (Table 1), and are ice-covered for > 50% of the year (SMHI 2006). The dominant land cover in the lake catchments is forests dominated by birch trees (*Betula* spp.) and Arctic heath (*Vaccinium* spp.,

Empetrum spp.), along with intermittent areas of bare rock (Supporting Information Table S1). Mean annual precipitation during 1960–1990 varied between 500 and 900 mm in the region, with > 45% as snow (https://www.smhi.se/).

Sampling

Sampling was conducted at the deepest point of each lake between June and October 2018 (open water), and in late March to early April (on safe ice before ice-melt) and in June 2019 (within 10 d after ice-melt). We collected water chemistry samples and measured physicochemical parameters in each season and deployed loggers for continuous data collection during the open-water period (detailed below). Depth profiles of temperature were measured every 0.5 m (0-4 m) or 1 m (4-10 m) using an optical sensor (ProDO; YSI). Photosynthetically active radiation (PAR) was measured every 0.5 m between 0 and 3 m using a LI-193 Spherical Quantum Sensor (LI-COR Environmental). The vertical light attenuation coefficient (K_d) of the 0–3 m water column was calculated as the slope between the natural logarithm of PAR against depth. We collected (Ruttner sampler, HannaNorden AB) water from 1 m beneath the lake surface or (in winter) beneath the bottom of the ice (representing epilimnion) and 1 m above the lake bottom (representing hypolimnion). For DIC, we injected 4 mL into airtight 20 mL borosilicate glass vials, pre-flushed with N₂ and containing 100 μ L 1.2 mol L⁻¹ HCl. Epilimnion and hypolimnion samples were pooled and subsampled for analysis of DOC (0.45 μ m filtered, acidified), absorbance at 440 nm (a_{440} , 0.45 μ m filtered), and total nitrogen (TN) and phosphorus (TP) at the Biogeochemical Analytical Facility at Umeå University following Verheijen et al. (2022).

Continuous data collection

As soon as possible after the ice breakup in 2018, a cup anemometer (S-WSA-M003, ONSET Corporation) was deployed 2 m above the ground < 15 m from the shoreline of each lake. The CO₂ concentration of the surface water at the deepest point of each lake was recorded hourly during the open-water season of 2018 using a floating chamber $(8 \times 8 \times 11 \text{ cm}^3)$ equipped with a CO₂-logger (K33 ELG; Senseair AB). The floating chamber was equipped with an internal drying chamber to prevent water from reaching the CO₂ sensor. Each hour, 150 cm³ of headspace was pumped from the chamber over a period of 30 s through a silica-filled drying chamber before reaching the CO₂ sensor and then looped back into the chamber. For further details of the chamber design, see Verheijen et al. (2022). Surface water temperature was measured every 5 min by a logger (Hobo TidbiT v2; ONSET Corporation) deployed next to the chamber. Continuous data collection of CO2 was also carried out in 2019 for 11 of the lakes, but these data are only used for the discussion of the timing of DIC sampling after the ice-off in 2019.

23782242, 2024, 2, Downloaded from https://aslopubs

h/doi/10.1002/lo12.10369 by Swedish University Of Agricultural

, Wiley Online Library on [26/03/2024]. See the Terms

Lake	Mean depth (m)	DOC (mg L ⁻¹)	<i>К</i> _d (m ⁻¹)	<i>a</i> ₄₄₀ (m ^{−1})	lce-melt evasion (g C m ⁻² yr ⁻¹)	Annual evasion (g C m ⁻² yr ⁻¹)
BD01	2.9	1.3	0.4	0.1	18.4	18.0
BD02	7.2	1.1	0.5	0.2	9.0	9.7
BD03	2.0	1.7	0.5	0.3	5.5	7.0
BD04	2.3	2.8	0.7	0.6	5.9	9.0
BD05	6.6	1.5	0.4	0.3	19.8	22.7
BD06	2.8	2.0	0.5	0.3	8.8	10.2
BD07	6.5	2.6	0.8	0.6	20.9	22.8
BD08	0.8	5.8	1.1	1.2	10.2	12.0
BD09	1.2	4.9	1.2	1.1	3.7	7.3
BD10	2.2	3.7	0.8	0.8	6.7	5.8
BD11	3.3	4.9	1.0	1.2	13.0	18.1
BD12	1.4	5.3	1.0	1.1	2.7	5.3
BD13	2.6	5.2	1.0	1.3	6.7	12.2
BD14	5.5	1.8	0.5	0.1	29.9	23.3

Table 1. Physical and chemical characteristics and CO₂ evasion of the 14 study lakes.

Lakes that were headwater lakes (i.e., had no lakes upstream) are marked in bold.

DOC, dissolved organic carbon; K_{d} , vertical light attenuation coefficient; a_{440} , absorbance at 440 nm.

Calculation of CO₂ fluxes

The methods for data handling and calculation of CO₂ fluxes are described in Supporting Information Text S2 and Verheijen et al. (2022). Briefly, CO_2 fluxes between the lake and atmosphere during the open-water season of 2018 (excluding the short ice-melt period) were estimated using Fick's first law of diffusion coupled with a wind-based model to determine piston velocity, k (Cole and Caraco 1998; Wanninkhof 2014). The total open-water CO₂ flux was calculated as the sum of all measured open-water CO₂ fluxes (excluding the days before the first DIC sampling following ice-melt). The CO₂ evasion at ice-melt 2019 was estimated as the difference between DIC inventory before and after ice-melt. We depth-integrated epi- and hypolimnion DIC concentrations over the respective volume of each layer to calculate DIC inventories. Epi- and hypolimnion volumes were determined using the function thermo.depth within the R package 'Rlakeanalyzer' (Read et al. 2011). We compared the results of density thresholds of 0.01–0.1 kg m⁻³ m⁻¹, yielding no differences in thermocline depths. Ice duration was inferred from satellite images (Supporting Information Text S1).

Statistics

We analyzed our data using R v 3.6.1 (R Core Development Team 2020). The data are archived on Zenodo (Karlsson et al. 2023). We tested for relationships between CO_2 evasion at ice-melt and in the open-water period, the proportion of annual CO_2 evaded during ice-melt, and their potential drivers using generalized least-squares linear regression analysis by means of the "gls" function in R. As family objects we used "Gaussian" with the "identity" link function for continuous variables and "binomial" with the "logit" link function for proportional variables. We log_{10} transformed data to conform with model assumptions, if necessary. We compared the proportion of annual flux evaded during ice-melt in relation to DOC concentration and mean depth in our study with previously published results from Arctic lakes. We compared the fits of single-variable models (DOC and mean depth) and of the model with both variables included (using the likelihood ratio test for nested models by means of the "Imtest" package in R). Two lakes were included in multiple studies, and in the statistical analyses, we included the data with the highest temporal resolution. We carried out the comparison with and without non-headwater lakes. The rationale is that the CO₂ evasion from non-headwater lakes is not only dependent on their intrinsic properties but also on the C cycling in, and export from, upstream lakes.

Results

The lakes generally had clear water (Table 1, average $K_d = 0.8 \text{ m}^{-1}$, $a_{440} = 0.7 \text{ m}^{-1}$), with low concentrations of DOC (Table 1, average 3.2 mg L⁻¹) and nutrients (TN = 140 µg L⁻¹, TP = 1 µg L⁻¹; Supporting Information Table S1). The lakes of this study are representative in terms of mean depth (Table 2; globally: 99% < 10.4 m; Cael et al. 2017), surface area (Arctic: 53% within 0.01–0.1 km²; Paltan et al. 2015), and DOC (Arctic: 72% < 7.4 mg L⁻¹; Klaus et al. 2021). A strong relationship between DOC and K_d ($R^2 = 0.93$, p < 0.01) suggests that water clarity and DOC content are predominantly controlled by variation in terrestrial colored DOC input, in line with previous findings showing a terrestrially dominated stable C isotopic

on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

		Ice-melt CO ₂ evasion				
Study	n	$g C m^{-2} yr^{-1}$	% of annual	DOC (mg L^{-1})	Mean depth (m)	Lake area (km ²)
This study	11 (14)	10.9 (11.5)	80 (79)	3.3 (3.2)	3.1 (3.4)	0.06 (0.06)
Karlsson et al. (2010)	1	8.1	28	7.2	2.0	0.02
Karlsson et al. (2013)	6 (12)	2.7 (4.1)	20 (43)	8.8 (9.1)	1.3 (1.4)	0.05 (0.05)
Jansen et al. (2019)	2 (3)	19 (28.5)	23 (21)	7.8 (7.8)	1.4 (1.5)	0.10 (0.07)
Preskienis et al. (2021)	2	40.1	61	4.4	2.5	0.07
Verheijen et al. (2022)	11 (14)	4.8 (5.9)	54 (57)	4.8 (5.0)	3.5 (3.4)	0.07 (0.07)
Total	33 (46)					
Average		9.6 (10.1)	55 (58)	5.2 (5.7)	2.7 (2.7)	0.06 (0.06)

Table 2. Mean ice-melt CO_2 evasion and its contribution (%) to annual CO_2 evasion, in this and previous studies of Arctic headwater lakes (values for all lakes, including both headwater and non-headwaters lakes, are given in parenthesis).

n Refers to the number of individual lakes sampled in each study. Jansen et al. (2019) do not include DOC, and for these lakes, we include the DOC data from Karlsson et al. (2013).



Fig. 1. The total CO_2 flux during ice-melt and open-water season. Positive values indicate flux from the lake to the atmosphere, and negative values indicate the reverse flux. The boxes indicate the upper and lower quartile, the horizontal line the median, the whiskers illustrate 1.5 times the interguartile range, and circles are outliers beyond the whiskers.

composition of DOC and low phytoplankton biomass in lakes in the region (Karlsson et al. 2003).

Lake CO₂ evasion was relatively high during ice-melt and relatively low (or in three cases negative) during the rest of the open-water season period (Fig. 1; Table 1). The average CO₂ evasion was 11.5 ± 7.9 g C m⁻² yr⁻¹ at ice-melt and 1.6 ± 3 g C m⁻² yr⁻¹ over the open-water period. Although some lakes were net CO₂ sinks during the open-water period, all were net sources of CO₂ to the atmosphere on an annual scale. The CO₂ evasion during ice-melt was positively related to log₁₀(mean depth) (generalized least-squares linear regression, $R^2 = 0.430$, p = 0.0151) but not related to DOC ($R^2 = 0.224$, p = 0.093). In partial contrast, the CO₂ flux



Fig. 2. The contribution of ice-melt period to the annual CO_2 evasion in Arctic lakes plotted against DOC concentration. Sources of published data are given in Table 2. The solid (headwater lakes) and hatched (all lakes) lines and shading represent the mean and 95% confidence interval of generalized linear model predictions.

during the open-water season was not related to mean depth ($R^2 = 0.08$, p = 0.34) or to DOC ($R^2 = 0.25$, p = 0.068).

The ice-melt evasion was comparable to previous studies of Arctic lakes, yet the degree to which it contributed to the annual CO₂ evasion was higher than previously reported ($80\% \pm 18\%$ vs. $47\% \pm 32\%$, Table 2). The proportion of annual CO₂ evaded during ice-melt for all lakes was negatively related to DOC ($R^2 = 0.42$, p = 0.011, Fig. 2) and positively related to log₁₀(mean depth) ($R^2 = 0.29$, p = 0.029). The relationships were stronger when excluding lakes with upstream lakes and only including headwater lakes (DOC: $R^2 = 0.63$, p = 0.009; log₁₀(mean depth): $R^2 = 0.37$, p = 0.038).

Karlsson et al.

Comparing the fits of single and dual variable models yields no improvement in adding mean depth to the DOC model (likelihood ratio test, $\chi^2 = 0.45$, p = 0.50), but a significant improvement of adding DOC to the mean depth model ($\chi^2 = 10.2$, p = 0.0014). This suggests that DOC is the superior explanatory variable. Given that the lakes experience similar climatic conditions, it is unlikely that variation in ice-cover duration (Supporting Information Text S1) has any major impact on the results.

Discussion

There were pronounced seasonal patterns in atmospheric CO2 exchange for most of the lakes, with particularly high CO₂ evasion during ice-melt in spring and relatively low and stable CO₂ fluxes during the rest of the open-water season (Fig. 1; Supporting Information Fig. S1). We did not observe any marked peaks in CO₂ evasion in autumn. Such peaks can be especially pronounced in lakes at lower latitudes as CO₂ accumulated in the hypolimnion during the summer is released by water column mixing during the autumn (Klaus et al. 2019). The lack of autumn peaks in our data likely reflects the relatively short period with stable thermal stratification and low net CO₂ production rates in the study lakes (Klaus et al. 2021; Klaus et al. 2022). The short episodic CO₂ evasion during ice-melt represented, on average, 80% of the annual evasion from these lakes. As three of the lakes were net CO₂ sinks during the open-water season (Table 1), neglecting the ice-melt evasion would erroneously classify them as CO₂ sinks.

The mass balance of DIC or CO₂ is commonly used for estimating ice-melt CO₂ evasion, yet it includes uncertainties. The DIC under ice was taken between late March and early April, as melting conditions later in the season caused difficulties in reaching the lakes and carrying out the sampling in a safe manner. This may have led to errors in calculated CO₂ evasion at ice-melt due to processes such as organic C mineralization and methane oxidation (resulting in underestimation) or photosynthetic CO₂ uptake (resulting in overestimation) under ice following sampling. Groundwater input may also cause errors depending on its magnitude and CO₂ content relative to lake volume and CO₂ content. We lack data to evaluate the net effect of these processes, which can be assumed to vary across lakes. Furthermore, the DIC after ice-melt was for most lakes taken within 10 d after the lakes became ice-free, assuming the entire column was equilibrated with the atmosphere. This is supported by pCO₂ having decreased by the time DIC was sampled (Supporting Information Fig. S2) and by calculated theoretical water column equilibration times (Z_{mean}/k) of on average 5.6 ± 3.9 d using the most conservative k model (Cole and Caraco 1998). Although this theoretical assessment contains uncertainties, it suggests that sampling within 10 d after ice-melt represents a compromise between maximizing the time to capture ice-melt

evasion and minimizing the time to avoid potential effects of other processes on the DIC mass balance. Still, part of the stored CO_2 under the ice is likely exported downstream rather than emitted following ice-melt. The uncertainties emphasize the need to assess the validity of the mass balance method, as well as to develop methods for direct measurements, for quantification of CO_2 evasion from Arctic lakes during ice-melt (Denfeld et al. 2018).

Since we do not have data on the potential CO₂ sources and sinks on an annual scale, we can only speculate on the drivers of the patterns in CO₂ exchange across lakes. The generally low DOC concentration implies relatively low net CO2 production in the lakes (Ask et al. 2012). Comparison with published data on net ecosystem production from the study lakes (Klaus et al. 2022) shows that internal CO₂ production during the open-water season is not sufficient to support annual CO₂ evasion (average 34% of CO₂ evasion) except for in two lakes. Although net ecosystem production during winter is unknown, it is reasonable to assume that external CO₂ input is an important contributor to CO₂ evasion from these relatively low DOC lakes. Regional stream studies generally show high stream water concentrations of DIC and CO₂ in winter and low concentrations in summer, while DOC exhibits the opposite pattern (Karlsson et al. 2013; Giesler et al. 2014). This suggests contrasting seasonal patterns in inorganic vs. organic C export from soils to lakes, which, in turn, could facilitate relatively high ice-melt: annual CO₂ evasion ratios from lakes.

It is also possible that primary production in the openwater season can play a role in the seasonal CO_2 exchange dynamics. While these lakes are nutrient-poor, favorable light conditions can maintain high growth rates by benthic algae that have access to nutrients in sediments (Hansson 1992; Vadeboncoeur et al. 2003). Published values of summer gross primary production (Klaus et al. 2022) in the lakes are similar to the CO_2 evasion at ice-melt presented in this paper. Although the generated autochthonous organic C is to some degree mineralized in the open-water season and stored in sediments, previous studies of high-latitude lakes show that autochthonous organic C generated in summer can be at least partly mineralized and contribute to the CO_2 accumulation in winter (Karlsson et al. 2008; Bogard et al. 2019).

Taken together, we hypothesize that in clear-water Arctic lakes, the importance of external CO_2 supply over internal mineralization of terrestrial organic C, and production and recycling of autochthonous organic C, both favor a high proportion of ice-melt period to annual CO_2 evasion. In lakes with higher DOC content, the internal mineralization of terrestrial organic C could play a larger overall role in CO_2 evasion, and this likely takes place predominately during the warmer open-water season with elevated DOC input, resulting in lower ice-melt : annual CO_2 evasion ratios. Furthermore, the CO_2 evasion at ice-melt was positively correlated to lake depth, in agreement with previous studies (Karlsson Karlsson et al.

et al. 2013; Ducharme-Riel et al. 2015). Although the model did not find a strong relationship between depth and the seasonal CO_2 dynamics in these lakes, depth likely contributed to the observed patterns with higher ice-melt : annual CO_2 evasion ratios in deep vs. shallow systems. Studies of the sources and control of CO_2 exchange in various lakes over the years are needed to improve our understanding of C cycling in Arctic lakes.

Based on the results, we suggest that CO₂ evasion during ice-melt is of particular importance for annual evasion in small clear-water Arctic lakes. As the study lakes were representative in size, depth, and DOC, the results should be applicable to a broad range of lakes in the Arctic and call for reassessment of CO₂ evasion from Arctic lakes. Yet, although the inverse relationship with DOC suggests underestimation of current CO₂ flux data based on open-water measurements in the abundant clear-water lakes, this effect is likely partly counteracted due to lakes with low DOC often having lower overall CO₂ evasion compared to lakes with high DOC (Raymond et al. 2013; Lundin et al. 2015). Irrespective, as the majority of global lakes are located at high latitudes and have a long ice-covered period (Denfeld et al. 2018), accounting for the high episodic CO₂ evasion is essential for global estimates of lake CO₂ evasion.

References

- Ask, J., J. Karlsson, and M. Jansson. 2012. Net ecosystem production in clear-water and brown-water lakes. Glob. Biogeochem. Cycles 26: GB1017. doi:10.1029/2010GB003951
- Bogard, M. J., and others. 2019. Negligible cycling of terrestrial carbon in many lakes of the arid circumpolar landscape. Nat. Geosci. **12**: 180–185. doi:10.1038/s41561-019 -0299-5
- Cael, B. B., A. J. Heathcote, and D. A. Seekell. 2017. The volume and mean depth of Earth's lakes. Geophys. Res. Lett. 44: 209–218. doi:10.1002/2016gl071378
- Cole, J. J., and N. F. Caraco. 1998. Atmospheric exchange of carbon dioxide in a low-wind oligotrophic lake measured by the addition of SF. Limnol. Oceanogr. 43: 647–656. doi: 10.4319/lo.1998.43.4.0647
- Denfeld, B. A., H. M. Baulch, P. A. del Giorgio, S. E. Hampton, and J. Karlsson. 2018. A synthesis of carbon dioxide and methane dynamics during the ice-covered period of northern lakes. Limnol. Oceanogr.: Lett. **3**: 117–131. doi:10. 1002/lol2.10079
- Ducharme-Riel, V., D. Vachon, P. A. del Giorgio, and Y. T. Prairie. 2015. The relative contribution of winter under-ice and summer hypolimnetic CO_2 accumulation to the annual CO_2 emissions from northern lakes. Ecosystems **18**: 547–559. doi:10.1007/s10021-015-9846-0
- Giesler, R., S. W. Lyon, C.-M. Mörth, J. Karlsson, E. M. Karlsson, E. J. Jantze, G. Destouni, and C. Humborg. 2014. Catchment-scale dissolved carbon concentrations and export

estimates across six subarctic streams in northern Sweden. Biogeosciences **11**: 525–537. doi:10.5194/bg-11-525-2014

- Hansson, L. A. 1992. Factors regulating periphytic algal biomass. Limnol. Oceanogr. 37: 322–328. doi:10.4319/lo.199 2.37.2.0322
- Jansen, J., B. F. Thornton, M. M. Jammet, M. Wik, A. Cortés, T. Friborg, S. MacIntyre, and P. M. Crill. 2019. Climatesensitive controls on large spring emissions of CH₄ and CO₂ from Northern Lakes. J. Geophys. Res.: Biogeosciences 124: 2379–2399. doi:10.1029/2019jg005094
- Karlsson, J., A. Jonsson, M. Meili, and M. Jansson. 2003. Control of zooplankton dependence on allochthonous organic carbon in humic and clear-water lakes in northern Sweden. Limnol. Oceanogr. 48: 269–276. doi:10.4319/lo.2003.48.1.0269
- Karlsson, J., J. Ask, and M. Jansson. 2008. Winter respiration of allochthonous and autochthonous organic carbon in a subarctic clear-water lake. Limnol. Oceanogr. 53: 948–954. doi:10.4319/lo.2008.53.3.0948
- Karlsson, J., and others. 2010. Quantifying the relative importance of lake emissions in the carbon budget of a subarctic catchment. J. Geophys. Res.: Biogeosciences **115**: G03006. doi:10.1029/2010jg001305
- Karlsson, J., R. Giesler, J. Persson, and E. Lundin. 2013. High emission of carbon dioxide and methane during ice thaw in high latitude lakes. Geophys. Res. Lett. **40**: 1123–1127. doi:10.1002/grl.50152
- Karlsson, J., H. A. Verheijen, D. A. Seekell, D. Vachon, and M. Klaus. 2023. Ice-melt and annual CO₂-evasion of arctic lakes [Data set]. Zenodo. doi:10.5281/zenodo.10080521
- Klaus, M., D. A. Seekell, W. Lidberg, and J. Karlsson. 2019. Evaluations of climate and land management effects on lake carbon cycling need to account temporal variability in CO_2 concentration. Global Biogeochem. Cycles **33**: 243– 265. doi:10.1029/2018gb005979
- Klaus, M., J. Karlsson, and D. Seekell. 2021. Tree line advance reduces mixing and oxygen concentrations in arctic-alpine lakes through wind sheltering and organic carbon supply. Glob. Change Biol. 27: 4238–4253. doi:10. 1111/gcb.15660
- Klaus, M., H. A. Verheijen, J. Karlsson, and D. A. Seekell. 2022. Depth and basin shape constrain ecosystem metabolism in lakes dominated by benthic primary producers. Limnol. Oceanogr. 67: 2763–2778. doi:10.1002/lno.12236
- Lundin, E. J., J. Klaminder, D. Bastviken, C. Olid, S. V. Hansson, and J. Karlsson. 2015. Large difference in carbon emission—burial balances between boreal and arctic lakes. Sci. Rep. **5**: 14248. doi:10.1038/srep14248
- Muster, S., and others. 2019. Size distributions of Arctic waterbodies reveal consistent relations in their statistical moments in space and time. Front. Earth Sci. **7**: 5. doi:10. 3389/feart.2019.00005
- Paltan, H., J. Dash, and M. Edwards. 2015. A refined mapping of Arctic lakes using Landsat imagery. Int. J. Remote Sens. 36: 5970–5982. doi:10.1080/01431161.2015.1110263

the applicable Creative

- Preskienis, V., I. Laurion, F. Bouchard, P. M. J. Douglas, M. F. Billett, D. Fortier, and X. Xu. 2021. Seasonal patterns in greenhouse gas emissions from lakes and ponds in a high Arctic polygonal landscape. Limnol. Oceanogr. 66: S117– S141. doi:10.1002/lno.11660
- Raymond, P. A., and others. 2013. Global carbon dioxide emissions from inland waters. Nature **503**: 355–359. doi: 10.1038/nature12760
- Read, J. S., D. Hamilton, I. Jones, K. Muraoka, R. Kroiss, C. Wu, and E. E. Gaiser. 2011. Derivation of lake mixing and stratification indices from high-resolution lake buoy data. Environ. Model. Software 26: 1325–1336. doi:10.1016/j. envsoft.2011.05.006
- Sobek, S., L. J. Tranvik, Y. T. Prairie, P. Kortelainen, and J. J. Cole. 2007. Patterns and regulation of dissolved organic carbon: An analysis of 7,500 widely distributed lakes. Limnol. Oceanogr. 52: 1208–1219. doi:10.4319/lo.2007.52. 3.1208
- Striegl, R. G., and C. M. Michmerhuizen. 1998. Hydrologic influence on methane and carbon dioxide dynamics at two north-central Minnesota lakes. Limnol. Oceanogr. 43: 1519–1529. doi:10.4319/lo.1998.43.7.1519
- Tranvik, L. J., and others. 2009. Lakes and reservoirs as regulators of carbon cycling and climate. Limnol. Oceanogr. **54**: 2298–2314. doi:10.4319/lo.2009.54.6_part_2.2298
- Vachon, D., C. T. Solomon, and P. A. del Giorgio. 2017. Reconstructing the seasonal dynamics and relative contribution of the major processes sustaining CO_2 emissions in northern lakes. Limnol. Oceanogr. **62**: 706–722. doi:10. 1002/lno.10454
- Vadeboncoeur, Y., E. Jeppesen, M. J. Vander Zanden, H. H. Schierup, K. Christoffersen, and D. M. Lodge. 2003. From

Greenland to green lakes: Cultural eutrophication and the loss of benthic pathways in lakes. Limnol. Oceanogr. **48**: 1408–1418. doi:10.4319/lo.2003.48.4.1408

- Verheijen, H. A., M. Klaus, D. A. Seekell, and J. Karlsson. 2022. Magnitude and origin of CO₂ evasion from high-latitude lakes. J. Geophys. Res.: Biogeosciences **127**: e2021JG006768. doi:10.1029/2021jg006768
- Verpoorter, C., T. Kutser, D. A. Seekell, and L. J. Tranvik. 2014. A global inventory of lakes based on high-resolution satellite imagery. Geophys. Res. Lett. **41**: 6396–6402. doi: 10.1002/2014gl060641
- Wanninkhof, R. 2014. Relationship between wind speed and gas exchange over the ocean revisited. Limnol. Oceanogr.: Methods **12**: 351–362. doi:10.4319/lom.2014.12.351
- Williamson, C. E., W. Dodds, T. K. Kratz, and M. A. Palmer. 2008. Lakes and streams as sentinels of environmental change in terrestrial and atmospheric processes. Front. Ecol. Environ. 6: 247–254. doi:10.1890/070140

Acknowledgments

We thank Karl Heuchel, Anders Jonsson, Sven Norman, Bror Holmgren, Petter Johansson, Collin Duinmeijer, and Bob Hendriks for their assistance in the field and laboratory, and the Biogeochemical Analytical Facility (BAF) at Umeå University for performing the water chemistry analyses. This study was financed by the Knut and Alice Wallenberg Foundation (Grant no. 2016.0083) and the Swedish Research Council (Grant no. 2016-05275).

> Submitted 21 July 2022 Revised 14 November 2023 Accepted 16 November 2023